Author's response to reviews

Title: Impact of an informed choice invitation on uptake of screening for diabetes in primary care (DICISION): Trial Protocol

Authors:

   Eleanor Mann (eleanor.mann@iop.kcl.ac.uk)
   A. Toby Prevost (toby.prevost@phpc.cam.ac.uk)
   Simon Griffin (simon.griffin@mrc-epid.cam.ac.uk)
   Ian Kellar (ik261@medschl.cam.ac.uk)
   Stephen Sutton (srs34@medschl.cam.ac.uk)
   Michael Parker (michael.parker@ethox.ox.ac.uk)
   Simon Sanderson (simon.sanderson@phgfoundation.org)
   Ann Louise Kinmonth (alk25@medschl.cam.ac.uk)
   Theresa M Marteau (theresa.marteau@kcl.ac.uk)

Version: 2 Date: 23 January 2009

Author's response to reviews:

The following changes have been made:

Affiliation for each author has been written in full in the following format: department, institute, city, and country.

Section headings have been formatted in sentence case.

The headings ‘Hypotheses’ and ‘Trial Registration’ and ‘Funding’ have been removed from the abstract, leaving only Background, Methods and Discussion.

Authors’ contributions have been formatted into a paragraph.

‘Additional files’ section has been deleted as there are no additional files attached to the manuscript (other than Figures).

Typographical errors have been corrected.